Title: Maintenance of bone mineral density after implantation of a femoral neck hip prosthesis

Authors:

Ralf Decking (ralf.decking@uni-ulm.de)
Christoph Rokahr (chrokahr@muenster.de)
Matthias Zurstegge (m.zurstegge@asklepios.com)
Ulrich Simon (ulrich.Simon@sfh-muenster.de)
Jens Decking (decking@uni-mainz.de)

Version: 4 Date: 10 January 2008

Author's response to reviews:

Tables- we included the table borders as single black lines.
Blank pages- All blank pages were deleted.
Figure order - The figure order was checked, Figures 3 and 4 were cited in the text as needed.